Author’s response to reviews

Title: The effects of gatekeeping on the quality of primary care in Guangdong Province, China: A cross-sectional study using Primary Care Assessment Tool-Adult Edition

Authors:

Cuiying Liang (liangcy3@mail2.sysu.edu.cn)
Jie Mei (mmjjj0926@163.com)
Yuan Liang (331891940@qq.com)
Ruwei Hu (huruwei@mail.sysu.edu.cn)
Li Li (lxl62@case.edu)
Li Kuang (kuangli@mail.sysu.edu.cn)

Version: 2 Date: 10 Jun 2019

Author’s response to reviews:

Dear Editor,

Thank you very much for your letter concerning our manuscript entitled “The effects of gatekeeping on the quality of primary care in Guangdong Province, China: A cross-sectional study using Primary Care Assessment Tool-Adult Edition” (FAMP-D-18-00318R2). We have checked the manuscript and provided figure legends under a separate heading of 'Figure Legends' after the References as required. We hope that the revised manuscript is now suitable for publication and look forward to hearing from you soon.

Yours sincerely,

Li Kuang, on behalf of all authors

Department of Health Administration
School of Public Health, Sun Yat-Sen University
No.74, Zhongshan Road #2, Guangzhou 510080, CHINA
Tel: 0086-20-87333687
Email: kuangli@mail.sysu.edu.cn